Author’s response to reviews

Title: Expanding the spectrum of A20 haploinsufficiency in two Chinese families: cases report

Authors:
Guomin Li (liguomin486@sina.com)
Hai-mei Liu (liu__hm@163.com)
Wan-zhen Guan (gwzleo@163.com)
Hong Xu (hxu@shmu.edu.cn)
Bing-bing Wu (bingbingwu2010@163.com)
Li Sun (lillysun@263.net)

Version: 5 Date: 12 Jun 2019

Author’s response to reviews:

RE: MGTC-D-19-00024R4, entitled "Expanding the spectrum of A20 haploinsufficiency in two Chinese families: cases report".

Dear editor:

Thank you for giving us a chance to revised our manuscript(MGTC-D-19-00024R4).

We read comments from reviewer and editor carefully, and have revised our manuscript based on their comments.

Best wishes

Children’s Hospital of Fudan University

Guo-min Li
Technical Comments:

- rename 'Acknowledge' to 'Acknowledgements'

We have renamed as Acknowledgements

Editor Comments:

Please address reviewer 3's request.

Reviewer reports:

Lara Bossini-Castillo (Reviewer 2): The authors have properly addressed my previous concerns.

Jennifer Below (Reviewer 3): When I requested genotype information on the pedigree, my intention was to request genotype information on all pedigree members who were genotyped, not only the mutation carriers. Sorry if I was not clear.

We have given genotype information to all pedigree members on Fig 1.